10/28/2009 CERES Data Management Team Meeting

Upcoming Events:

Science Team Meeting – November 3 – 5 in Ft. Collins, CO

HDF5 Workshop – November 3 – 5 in Maryland near Goddard

Notes from the previous meeting

(Held 10/14/2009 at 10:00 a.m. in Room 2316 at the ASDC)

ANNOUNCEMENTS

If you have any computer concerns please provide them to Walt.

For SA support,

- 1. If you are sure that you have an ODIN issue, send a help ticket to SDdesktop@asdsun.larc.nasa.gov.
- 2. If you have a non-ODIN issue OR are unsure that it is an ODIN issue, send a help ticket to help@asdsun.larc.nasa.gov.
- 3. If you don't hear back from help@asdsun.larc.nasa.gov within 1-2 days, send email to Caroline Rasmussen and include the original help ticket.
- 4. If you have an ODIN password issue, call x47777.

Please use the DMT alias (ceres-dmt@lists.nasa.gov) for emails that pertain to the entire DMT. There is also a DMTM alias (ceres-dmtm@lists.nasa.gov) that includes those who attend the DMTM.

Please note if activity is for Edition2 or Edition3 in status reports.

Raise visibility of new files as soon as possible so changes can be made to ANGe. A few weeks notice would be nice; put it in your DMTM status report.

ASDC and **SCF**

SIT:

The following are in SSI&T: GGEO 653, Instantaneous SARB 718, Clouds 719, and Synoptic SARB 721.

Production:

No updates to report.

ValR14 is still being evaluated.

From the previous report: Dale asked if there was any overlap for Snow comparison. There isn't, but Lisa can submit a PR requesting it. *Today's update*: Sue will run overlap month.

User Services:

Updated tables to reflect current product dates.

SCF:

Denise reported that the leapsec and utcpole files haven't been updated on the SCF machines since January. She asked that they be automatically updated on the SCF machines like they are on other machines.

Sunny would like a way to identify which MODIS files are used for various runs; perhaps this should go to the SEC.

The Toolkit on AMI was moved from ASDC_archive to SPG_ops.

Quotas for *AMI* will be published on a Web site; the DMT will be notified when the information is available.

PRs

For PRs 65-09 and 64-09 Denise pointed out that the ending Data Dates should be 12/01/2004 for each rather than 05/01/2005.

\mathbf{CM}

Discussed Instantaneous SARB SCCR 72 and ERBE-like SCCR 729 – no problems.

Discussed schedule updates.

SEC

No updates.

Toolkit

No updates.

SUBSYSTEM STATUS:

Instrument

Delivered PGE CER1.2P1 to CM for AMI.

Evaluating ValR13.

Sue – TRMM DPREP needs to get off *warlock*.

ERBE-like

Edition3 gains and spectral response function going well.

Clouds

Preparing for Science Team Meeting.

Evaluating ValR14.

Convolution

From Sunny - Evaluating ValR14; not too happy with the results; trying to get a better overall picture.

Inversion

All SOFA algorithms are installed; now evaluating them.

SARB

Ported PGE CER5.4P2 to *magneto* and new directory structure; will be porting PGE CER5.1P2 next and delivering to CM on October 30.

Tom needs to know if anyone still uses water vapor data from SSMI; if not, ASDC will no longer need to get this data.

TISA Averaging

Tried to evaluate GGEO – no luck.

Ran March 2000 - December 2007 through SRBAVG-Lite.

Working on Edition3 DPC.

NPP Issues

GSIT is scheduled for November 17 – December 10.

Jim is planning to attend the GIST Test Datga Summit, Oct 21-22 in Suitland.

Travel for FM5 requires Mike C. approval.

Meeting ended at 10:58 a.m.